Author's response to reviews

Title: Suicide attempt by self-stabbing of the bladder: a case report

Authors:

Youssef Kharbach (dr.kharbach@gmail.com)
Somuah Tenkorang (snrsomuahtenkorang@gmail.com)
Mustapha Ahsaini (drahsaini@gmail.com)
Soufiane Mellas (drsmellas@yahoo.fr)
Jalal Eddine El Ammari (jeelammari@yahoo.fr)
Mohammed Fadl Tazi (mfadltazi@gmail.com)
Abdelhak Khallouk (drakhllouk@yahoo.fr)
Mohammed Jamal El Fassi (mjelfassi@hotmail.fr)
Moulay Hassan FARIH (mhfarih@hotmail.fr)

Version: 2
Date: 13 August 2014

Author's response to reviews:

Dear Editor,

It is an honor to us to submit our manuscript to your relevant journal.

We report the case of a 56-year-old Moroccan male, known to be suffering from an untreated paranoid schizophrenia with history of several previous suicide attempts, who was presented to the Emergency unit for self stabbing with a 15 cm sewing needle at the hypogastric region of the abdomen with full penetration of the needle in the abdomen in suicide perspective.

Psychiatric disorders have been reported in patients admitted for self-introduction of foreign bodies in the bladder as an act of sexual satisfaction whereas very few cases have been reported in the English literature for the same practice in the context of suicide.

Kind regards